Author's response to reviews

Title: Collaborative Care for patients with severe borderline and NOS personality disorders: a comparative multiple case study on processes and outcomes.

Authors:

Barbara Stringer (b.stringer@ggzingeest.nl)
Berno van Meijel (berno.vanmeijel@inholland.nl)
Bauke Koekkoek (b.koekkoek@propersona.nl)
Ad Kerkhof (ajfm.kerkhof@psy.vu.nl)
Aartjan Beekman (a.beekman@ggzingeest.nl)

Version: 2 Date: 30 May 2011

Author's response to reviews: see over
30 May 2011

The BioMed Central Editorial Team

BMC Editorial Office

Dear Editor,

Please, find enclosed our updated manuscript, entitled ‘Collaborative Care for patients with severe borderline and NOS personality disorders: a comparative multiple case study on processes and outcomes’, which you decided, in principle, to publish in BMC Psychiatry as a study protocol.

The reviewer required no major or minor revisions. In the updated manuscript the trial registration number is added in the abstract. We checked the manuscript with the formatting checklist.

The study has been approved by the Medical Ethical Committee of VU University Medical Center, Amsterdam, the Netherlands. The study is funded by VU University Medical Center and GGZ inGeest, Amsterdam, the Netherlands, and INHolland University for Applied Research Amsterdam, the Netherlands.

The manuscript is original, has not been published in any other journals, nor is it currently under consideration by another journal. All authors have participated in the design of the study and we take responsibility for the integrity of the manuscript. All authors read and approved the final manuscript.

Yours sincerely

Barbara Stringer